Author’s response to reviews

Title: Low dose of flurbiprofen axetil decrease the rate of acute kidney injury after operation: a retrospective clinical data analysis of 9915 cases

Authors:

Dong Wang (905486084@qq.com)
Shi-Kun Yang (yangshikun@csu.edu.cn)
Meng-Xi Zhao (249619441@qq.com)
Yong-Zhong Tang (tangyongzong@csu.edu.cn)
Wen OU-Yang (ouyangwen133@vip.sina.com)
Hao Zhang (zanghaoliaoqing@163.com)
Qin Liao (xy3yiliaoqin@sina.com)

Version: 3 Date: 27 Jan 2020

Author’s response to reviews:

Dear Dr Charat Thongprayoon,

We respectfully send back our manuscript: Low dose of flurbiprofen axetil decrease the rate of acute kidney injury after operation: a retrospective clinical data analysis of 9915 cases (Manuscript-ID: BNEP-D-19-00829.R3) for the consideration of publication in the Journal of BMC Nephrology.

We greatly appreciate the efforts of the Editors for the valuable and constructive comments on previous draft. Based on their comments and requests, we have carefully taken their comments into consideration and made extensive modification on the original manuscript. The following summarizes how we responded to reviewer comments. A revised manuscript with the correction sections marked was attached as the supplemental material and for the purpose of easy check.

Here is our response to their comments.

Revision — authors’ response

Editor Comments:
Comment-1:
Abbreviations
1. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.
Response: Thanks for your valuable comments. We have added a list of abbreviations before the Declarations section, please check it, thanks a lot.

Comment-2:
2. Consent for publication
Currently, the statement in your “consent for publication” section of your declarations is incorrect. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.
Response: Thanks for your valuable comments, we have amended relevant content, please check it, thanks a lot.

Comment-3:
3. Medical records permissions
Thank you for including your Ethics approval statement in the Declarations. We notice your study is a retrospective study and that you state that informed consent was waived. Please clarify if any further permission from the hospital was required to access the medical records of patients.
Response: Thanks for your valuable comments, This study was in line with the guidelines of the Strengthening of Observational Epidemiological Studies (STROBE) statement, in accordance with the Helsinki declaration, and approved by the ethics committee of the third Xiangya hospital of Central South University (2018-S376). In order to access the medical records of patients, we need the permission of Big data analysis center of the Third Xiangya hospital of Central South University.

Comment-4:
4. Clean copy
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: Thanks for your valuable comments, we have uploaded a clean version of revised manuscript, please check it, thanks a lot.

We will give our heartfelt thanks for the reviewers’ and the editors’ comments, and we believe that the amendments will make this manuscript more valuable. We hope the reviewers and the editors will be satisfied with the revisions for the original manuscript. If you have any question about this paper, please don’t hesitate to let us know.
Thanks and best regards!

Yours sincerely,
Dr. Qin Liao,

Department of Anesthesiology
The Third Xiangya Hospital, Central South University
No.138, Tongzipo Road, Changsha, Hunan Province, China
Jan 27, 2020